Author's response to reviews

Title: Decompressive cervical Laminectomy and Lateral Mass Screw-rod arthrodesis. Surgical Analysis and outcome.

Authors:

Moh’d M Al Barbarawi (dr_barbarawi@yahoo.com)
Ziad A Audat (ziadaudat@yahoo.com)
Moutasem M Obeidat (obeidat74@yahoo.com)
Tareq M Qudsieh (dr_tareqqod@yahoo.com)
Waleed F Dabbas (waleeddabbas@yahoo.com)
Mouness H Obaidat (mouness1@hotmail.com)
Anas A Malkawi (asmalkawi1984@yahoo.com)

Version: 5 Date: 17 May 2011

Author's response to reviews:

the editorial team,

dear all.

i went through the new comments and revised the manuscript as follows:

1- the names written correctly in both manuscript and the system

2- all co-authors have reviewed and signed the finally edited manuscript.

3- the signitures of co authors is added as a separate page in the end of the manuscript.

4- the shades on tables were taken out.

4-typography was reviewed and approved the way it is and no corrections done.

with this revision we hope it is ready for publication.

your acceptance for publication is appreciated.

thanks a lot.

dr barbarawi